Author's response to reviews

Title: Growth in Children with Cerebral Palsy during five years after Selective Dorsal Rhizotomy: a practice-based study

Authors:

Lena Westbom (lena.westbom@med.lu.se)
Annika Lundkvist Josenby (annika.lundkvist@med.lu.se)
Philippe Wagner (pw@nko.se)
Eva Nordmark (eva.nordmark@med.lu.se)

Version: 5 Date: 30 June 2010

Author's response to reviews:

To the editorial production team
Figures OK!
Made two changes in the main text:
page 11 par 2 row 5 "Figure 9" should read Figure 6
page 25 in Table 1:
30,3% should read 30%; 26,8% should read 27%
My fault...
Best wishes,
Lena Westbom